vc/2-Oct-03
Week 40 starting 29 Sept 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

 Today(30/09/03)
	Forward look,   & Next Magnet 

	TBA1
	3047
	Should be started cooling down today  
	Cold Tests Start Thurs ?After this magnet, Bench to be de-activated – preparation for D bench

	TBA2
	3042
	cold
	 Cold Tests StOP Thurs morn?

	TBC1
	2021
	Need to remove today
	Next Magnet 3045; Connect & Cooldown by Fri. midday; Start Cold Tests on Fri  during w’end

	TBC2
	SSS1
	cold
	Cold tests

	TBE1
	2002
	Finish Connection ; Should be started cooling down today  
	COOLDOWN START ; Cold Tests Start Thurs ?

	TBE2
	2010 
	Need to remove today
	Next Magnet 3039;

	TBF1
	2025 after F2 done? 
	
	Prepare For 

Commissioning AFTER F2

	TBF2
	2025 proposed
	
	Prepare For 

Commissioning FIRST


Test Coordinator this week:  S.Sanfilippo;  MTM  Technician on call:  (16-3602) : Federico

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
· Problems from Operation Last week (Stephane + Test Supervisor )
	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 23_A


	TBA2 :IAP@warm : Rheaters 25 ohms
	MTM
	IAP
	Incorrect coefficients

	Tuesday 23_A


	TBA2:LF Acquisition : First led off but message acquisition armed
	MTM
	LF
	PC reset and OK

	Tuesday 23_A


	TBA2 : HV test could not be launched because cable on passive bus bar was broken
	MTM
	
	Cable from TBA1 taken. 

Repaired by Richard.

	Tuesday 23_A


	TBA2 : HV test could not be launched because power supply current leads connected
	MTM
	
	Test coordinator has to plan the disconnection before 

	Thursday 25_M


	Mobile rack for HV test used in C1
	MTM
	
	Connection cables for passive burs bar broken.

Richard was not informed 

	Thursday 25_A

Friday 26_A


	HV test at warm (TBE) capacitance problem.
	MTM
	HV
	Equip.support called.

New version of soft  not ready.

	Thursday 25_N


	TBA2 : Problem with T anticryostat
	MTM
	
	Equip.support called.

Resolved.

	Friday 26_A
	Shaft 21 (TBC1) : S02 coil e open.
	MTM
	
	Trace to be kept.

	Friday 26_A


	TBE2: Min energy at cold

LF config not found
	MTM
	LF
	????

	Friday 26_N


	TBC1: Provoked quench all LF

Faulty quench in bus bar
	AB/CO
	AQA
	No MIITS, No current , Origin of Quench +Vext

Saturday : Nicolai decide to continue the training.

Software problem on investigation.

	Friday 26_N


	TBA2: HV test 

Dipole–QH vs Ground >limit
	MTM
	
	OK the morning after.

Humidity problem in leads?

	Saturday  27_N


	TBA2: IAP at cold  D1-L-U (144 mV)

AC compensation


	MTM
	IAP
	Tuning of the channel?

	Sunday 28_A


	TBA2: Training quench 2 : Wrong extension file given


	MTM
	
	Extension .a02

	Sunday 28_N


	TBA2: Motor problem during full load line

MM on TBA2: Field polarity error on ap2
	MTM
	
	Aborted and re-launched.

Check carefully the cases explained in the table.

	Monday 29_A


	TBA12: TEMA not working for IAP at warm
	AB/CO?
	TEMA
	TEMA launched in local

OK

	Monday 29_A


	TBA2: Power supply did not start.

Magnet shorted to the ground.
	MTM
	
	Cables coming from the patch panel used for RRR removed and OK.


